

Essex Finance CentreWorking Paper Series

Working Paper No 69: 03-2021

"Trade Credit and Firm Investments: Empirical Evidence from Italian Cooperative Banks"

"Stefano Filomeni, Michele Modina, Elena Tabacco"

Trade Credit and Firm Investments: Empirical Evidence from Italian Cooperative Banks

Stefano Filomeni, Michele Modina, and Elena Tabacco*

March 18, 2021

Abstract

By exploiting a unique and proprietary panel dataset comprising 18,682 Italian SMEs operating with 99 cooperative banks over the period 2008-2014, we investigate the influence of the trade credit channel on firm investment decisions in the Italian market, distinguished by a considerable presence of relationship cooperative banks' branches with a heterogeneous geographical distribution. Firstly, our findings confirm a significant influence of the trade credit channel on firm investment decisions. Secondly, we document that those SMEs located in those Italian provinces with an abundance of cooperative banks' branches rely less on trade credit to finance investments. Lastly, we show that longer firm-bank relationships decrease firm dependence on trade credit to boost investments. Our study is of particular relevance because it strengthens the effectiveness of the commercial credit channel for SMEs in spurring corporate investments. Indeed, fostering a deep understanding of the real effects of firm financing sources is paramount to encourage investment by SMEs and to allow them to preserve their positioning in the market. Moreover, we exploit the Italian market, well-suited to perform such an analysis, since it is characterized by more inter-personal financing relationships as compared to other countries.

Keywords: SMEs, trade credit, investment, relationship lending, soft information, cooperative bank

JEL codes: G21, D22, D82

-

^{*} Stefano Filomeni (corresponding author), stefano.filomeni@essex.ac.uk, University of Essex, Essex Business School, Finance Group (Colchester, UK); Michele Modina, michele.modina@unimol.it, University of Molise (Campobasso, Italy); Elena Tabacco, elena.tabacco@yahoo.it, BNP Paribas Leasing Solutions (Milan, Italy). The opinions expressed herein are those of the authors and do not necessarily reflect those of BNP Paribas Leasing Solutions or any affiliated entities of the BNP Paribas Group.

1. Introduction

In a perfect capital market, trade credit investment and financing decisions are independent because companies enjoy an unlimited access to a wide array of sources of finance in a scenario in which asymmetric information is absent (Modigliani and Miller, 1958): companies are always capable to obtain external finance without problems at a reasonable price. However, in a more realistic imperfect capital market, companies may face an opportunity cost associated with trade credit due to the potential difficulty experienced by a firm to access credit through different sources of financing, such as bank credit. This is particularly relevant for small and midsize enterprises (SMEs) for which this opportunity cost is likely to be higher due to their informational opaqueness. Indeed, SMEs have borrowing issues related to information asymmetries and mostly rely upon commercial trade and relationship lending based on soft information. This feature not only makes trade credit the only other viable external source to bank credit during their life cycle (Canto-Cuevas et al., 2019), but also raises the likelihood that SMEs will get bank credit-constrained due to greater perceived risk. In this scenario, creditconstrained SME are oriented to turn to trade credit which is the most important alternative to bank lending as a source of external financing (Carbo'-Valverde et al., 2016) in nearly every economy (Demirgüç-Kunt and Maksimovic, 2002)¹. In contrast, larger firms are less likely to fall credit-constrained as they tend to borrow from large financial institutions that mostly rely on hard information in their lending decisions (Filomeni et al., 2020a, 2020b; Stein, 2002; Berger and Udell, 2006). In this regard, Berger and Udell (2006) show that large institutions tend to lend to larger firms, while small institutions lend more to smaller firms based on stronger bank-firm relationships (Haynes et al., 1999; Cole et al., 2004; Scott, 2004; Berger et al., 2005b).

Within this context, this paper provides a novel contribution to the literature by investigating the extent to which SME reliance on the trade credit channel to finance investment decisions is affected by the structure of the local banking system and relationship banking features. To address our research objectives, on the one hand, we focus on SMEs because SMEs represent the backbone of the Italian economy and typically have fewer options to access capital markets as they are less transparent than their larger corporate counterparts and poses a higher credit risk (Petersen and Rajan 1997; Carey et al., 1993; Berger and Udell, 1998). Trade credit plays an important role in SME financing decisions (Hernandez-Canovas and Martinez-Solano, 2007; Ogawa et al., 2013; Martinez-Sola et al., 2014) and it is the only major source of financing

_

¹ In the US trade credit provides as much external finance as bank loans representing approximately one third of the debt of US SMEs (Berger and Udell, 2006). In the sample used by McGuinness et al. (2018), the more than two hundred thousand companies, located in 13 European countries, are net suppliers of 62 billion euros of trade credit, equal to 14% of total assets.

(Berger and Udell, 1998)². On the other hand, Italy is characterized by geographical heterogeneity in the local banking system across the country: several provinces have an abundance of cooperative banks (Alessandrini and Zazzaro, 1999; Alessandrini et al., 2009), mostly relying on soft information in their credit relationships. Indeed, this allows us to exploit within-country geographical variation in the degree of inter-personal financing relationship characterizing the local banking system. At the same time, Italy provides an ideal setting for our study as SMEs represent a large segment of the corporate market (Bonaccorsi di Patti and Gobbi, 2001; Guiso et al., 2004; Benfratello et al., 2008; Alessandrini et al., 2009; La Rocca et al., 2010). Therefore, the Italian capital system, characterized by more inter-personal financing relationships compared to other countries, is well-suited to perform our analysis.

In order to address our aforementioned research objectives, firstly, we investigate the real effects of the trade credit channel on long-term investments (Carbò-Valverde et al., 2016; Ferrando and Wolski, 2018). We do so by using a panel dataset comprising 18,682 SMEs operating in Italy in the time period 2008-2014. Following Love et al. (2007), Goncalves et al. (2018), and D'Mello and Toscano (2020), we measure trade credit as the net effect of extended and received trade credit, i.e., net trade credit³. Secondly, we specifically explore whether the local banking system affects the relationship between trade credit and SME investment decisions. We do this by exploiting within-country variation in the different degrees of inter-personal financing relationship of the local banking system, characterized by a heterogeneous geographical distribution of relationship cooperative banks' branches. Thirdly, the granularity of the data at our disposal allows for the collection of information not only on firm accounting data, but also on specific firm-bank relationship features. In this regard, we study the differential effects of trade credit on firm investments by accounting for firm-bank relationship heterogeneity based on the length of the credit relationship and on the bank-borrower distance. Indeed, longer and closer firm-bank relationships lead to an increased collection of proprietary "soft" information on the part of the bank following repeated interactions with the same firm over time (Petersen and Rajan, 1994; Berger and Udell, 1995; Uchida et al., 2012; Bolton et al., 2016; Beck et al., 2018). Overall, we believe addressing these questions is of crucial importance for policy makers in order to draw appropriate conclusions on the real effects of the trade credit channel as a

_

² For Italian firms, trade credit is a very important source of finance (e.g., Agostino and Trivieri 2014): during 2003–2009, accounts payable constituted, on average, 19.6 % of total assets, while accounts receivable constituted 29.2 % (source: Bank for the Accounts of Companies Harmonised, available at http://www.bachesd.banque-france.fr.).

³ This allows us to take into account the fact that firms manage trade receivables and trade payables concurrently to optimise performance (Ferrando and Mulier, 2013) and manage risk (Fabbri and Klapper, 2008) since these two components of trade credit exert influence on each other (Niskanen and Niskanen, 2000). That is, firms are both trade creditors and trade debtors at the same time (Burkart and Ellingsen, 2004) and, thus, simultaneously manage receivables and payables (Hill et al., 2010).

financing source alternative to bank lending, and on whether the magnitude of the trade credit effect differs according to specific features of the bank-firm relationship and the local banking system. To the best of our knowledge, this study represents the first attempt to investigate the real effects of the trade channel for SMEs' investment decisions by exploiting variation in interpersonal financing relationships characterizing the different geographies of the Italian banking market. Moreover, the specificity of the sample, which refers to small businesses financed by local banks, allows us to investigate the phenomenon at a much higher level of detail than it is typically found in the literature.

By way of preview, firstly we find a significant influence of the trade credit channel on firm investment decisions, suggesting that net trade credit significantly affects the growth rate of firm investment. Indeed, an increase in net trade credit triggers a decrease in the year-on-year percentage change of firm investments. This negative effect can be attributed to the liquidity-absorbing consequence of an increase in net trade credit. Secondly, we document that those SMEs located in Italian geographical areas characterized by an abundance of cooperative banks' branches, relying on soft information-intensive relationship banking, rely less on trade credit to finance their investment decisions. This is supportive of Berger et al. (2005b)'s view that local cooperative banks still benefit from a competitive advantage over nationwide banks in small business lending, since the latter are characterized by organizational complexity and face more severe communication frictions due to the greater distance between their headquarters and local branches. Lastly, we provide evidence that shorter firm-bank relationships lead to a greater dependence of companies on the trade credit channel to boost investments, while firm-bank geographical proximity does not exert a significant influence on this nexus due to the nature of our sample of cooperative banks operating with SMEs mostly on a local basis.

Within the related literature, our paper is closely related to Carbò-Valverde et al. (2016), hereafter referred to as C-V, who analyze for the first time whether commercial credit provided an alternative source of external financing for SMEs' investments during the 2008 crisis by using financial and banking data of nearly 40,000 businesses in Spain over the period 1994-2010. Similarly to C-V's work, our paper assesses whether bank credit-constrained SMEs turned to trade credit as an alternative source of external financing to boost investments using information on firm characteristics and bank-firm lending relationships. However, our paper differs from C-V's study by adding new dimensions to the analysis related to relationship banking: it exploits within-country variation in inter-personal financing relationships characterizing the different geographies of the Italian banking market.

Our paper is also close to Ferrando and Wolski (2018) who investigate the relationship between net trade credit and firms' investment levels by focusing on financially distressed firms. They provide evidence that, while net trade credit has an overall negative impact on investment level due to its liquidity-absorbing effect, this effect is less pronounced for more financially distressed firms since, through capital expenditure, the latter aim at maintaining crucial business relations with their customers in order to participate in the final profits via trade credit repayments. However, even if our results are supportive of the notion that a firm positive net trade credit position is liquidity-absorbing as it reduces the firm growth rate of investments, our paper extends the analysis by contemplating the relevance of the specific features of the bank-firm relationship within the context of trade credit. Moreover, we observe the exact firm-bank relationship not just relying on a "matching procedure based on a fuzzy-matching algorithm subject to potential biases stemming from spurious firm-bank relations" (Ferrando and Wolski, 2018).

Moreover, we add a relationship banking dimension (Giannetti et al., 2011) and focus on bank credit constrained SMEs mostly relying on the trade credit channel for their investment choices (Carbò-Valverde et al., 2016). Unlike the focus of Giannetti et al. (2011) on commercial credit and on aspects of bank-firm relationships in the United States, the Italian dimension of our analysis allows us to investigate the impact of commercial credit on investment decisions in a more bank-oriented economic context.

With the crisis generated by Covid-19, our study takes on even greater relevance. The shock that affected the economic world due to Covid-19 was strong. With countries accounting for more than 50% of world GDP frozen for at least two months, the decline in revenues has been more sustained than in previous recessions. The most vulnerable companies are those of smaller size that do not make use of securities markets and for which recourse to government funds is not always easy. Without considering the effects of government support interventions, there is a strong and ongoing concern that SMEs have been hardly hit by the pandemic. In this context, finding adequate solutions to help SME's is essential given that over two thirds (70%) of global employment is provided by small economic units (ILO, 2019). Understanding the effects of trade credit and its function of supporting investment activity in times of crisis can encourage academics, policy makers and operators to find the most appropriate solutions (e.g. supply chain credit) to support SMEs in absorbing the impact of the current crisis and continue investing even in uncertain scenarios.

The remainder of this paper is organized as follows. Section 2 presents the Italian banking system with a focus on its geographical heterogeneity, relationship banking, and

cooperative banks, Section 3 discusses the relevant literature, Section 4 describes our data, Section 5 presents our empirical methodology, Section 6 highlights the results of our empirical analysis, Section 7 reports on several robustness tests and, last, in Section 8, we conclude.

2. The Italian banking system: cooperative banks, relationship banking and geographical heterogeneity

Cooperative banks play an important role in the capital markets of many countries (McKillop et al., 2020)⁴. In Italy, today they are the most representative form of banking localism. Cooperative credit banks have a direct presence in more than a third of the Italian municipalities and, in 620 municipalities (out of 7,903), they operate as a single intermediary (Bank of Italy, 2019). Their vocation to retail banking is evidenced by the fact that 59% of assets are destined for loans to households and small- and medium-sized businesses (6% more than other banks). Support for the territory is confirmed by the destination of the collected resources: for every 100 euros of savings collected in the area, 87 euros become credit to the real economy of that area.

The small size and the orientation to the local market favour relationship lending and the reduction of information asymmetries between lenders and borrowers (Petersen and Rajan, 1994; Berger and Udell, 1995; Berger et al., 2004; Elsas, 2005; Beck et al., 2018). The information advantage enjoyed by cooperative banks over their larger counterparts, as well as their proximity to the entrepreneurial and social fabric of the territory, translate into a better capacity to select and monitor opaque borrowers such as SMEs (McKillop et al. 2020). The superior ability to collect and manage information relating to customers, especially soft information, has meant that the impact of the financial crisis on the availability of credit disbursed by cooperative banks was less severe than the one observed in different types of banks (Ferri et al., 2014). Indeed, in the last decade the cooperative banks introduced net loans by over €11 billion into the economic circuit with an increase of 9.1% higher than the overall growth in the credit market of +4.6% (Lopez et al., 2019).

While Italy has been unified for the last 150 years, the local banking system varies notably across provinces. As a matter of fact, several provinces are characterized by an abundance of small cooperative banks that operate in restricted territorial areas (Alessandrini and Zazzaro, 1999; Alessandrini et al., 2009). Local cooperative banks benefit from competitive advantages over nationwide banks, as the latter are afflicted by organizational complexity and more severe problems in communicating information (Berger et al., 2005b; Filomeni et al., 2020a). Indeed,

6

⁴ Cooperative banks are a key component of the cooperative movement in the credit sector, which originated in Europe in the nineteenth century as a response to the problems that small urban and rural businesses had in obtaining credit.

the distance between the bank headquarters and local branches of nationwide banks gives rise to information frictions within the banking organization, because the bank headquarters are less able to interpret the information coming from distant branches than information from closer ones (Stein, 2002; Liberti and Petersen, 2019; Filomeni et al, 2020a; Filomeni et al, 2020b).

On the one hand, while the lending decisions of large national banks tend to be based on hard information, small cooperative banks rely on soft information collected directly and indirectly through personal bank-firm relationships and continuous interaction with local firms (Howorth and Moro, 2006). On the other hand, suppliers provide credit to their customers because the soft information accumulated in repeated trading relationships provides them with a significant advantage over banks in granting credit (Biais and Gollier, 1997; Petersen and Rajan, 1997). It follows that, if both firm trade credit decisions and local cooperative banks' credit decisions rely on soft information, trade credit and cooperative bank loans could act as substitutes. Consistent with this view, a high proportion of cooperative bank branches in a province should reduce the need for trade credit and, as such, affect the relationship between trade credit and firm investment decisions. As a matter of fact, a local bank, whose employees are part of the local community, and that may be owned or managed by local community members, possesses a more direct and in-depth knowledge of firms located in its operating area. Indeed, the local bank participates to the local community life, thus collecting information not available to banks that operate at a distance (Angelini et al., 1998; Stein, 2002). Moreover, even if nationwide banks' local branches may integrate borrowers' hard information with valuable soft information collected locally, or if large complex banks use transaction lending technologies wellsuited to SMEs such as credit scoring (Berger and Udell, 2006; Ferri and Neuberger, 2014), community banks are still expected to benefit from an informational advantage in providing loans to local firms due to their engagement in relationship lending (Bolton et al. 2016; Filomeni et al., 2020a; Filomeni et al., 2020b). Bartoli et al. (2013) note that transactional lending, even when using sophisticated technologies, does not substitute for relationship banking in the granting of soft-information intensive loans to SMEs. Presbitero and Zazzaro (2011) prove evidence that the organizational structure of local credit markets influences relationship lending. On the one hand, in markets where large, out-of-market banks predominate, increases in interbank competition are detrimental to relationship lending. On the other hand, in markets where a large group of small local banks operate, increased competitive pressure from outside can pushes banks to further cultivate their relationship ties with customers.

3. Literature review and hypotheses development

The relevance of trade credit as an important alternative source to bank finance has not gone unnoticed within the academic community, where several papers have analysed both the financial and economic aspects of trade credit (Pattnaik, 2020a) and the multi-disciplinary nature of interfirm credit transactions (Pattnaik, 2020b). In this regard, Pattnaik et al. (2020a) and Pattnaik et al. (2020b) provide a comprehensive and up-to-date overview of the trade credit literature. While the former focuses on the financial and economic aspects of this literature, the latter displays a multidisciplinary nature in line with the evolution and growth of related research. Paul and Boden (2008) recognize that the theoretical and empirical exposure of the commercial credit literature is diversified with contributions from multiple streams of studies in order to better investigate the numerous factors influencing the trade credit channel.

From an industrial perspective, product natures (Giannetti et al., 2011), buyer-supplier relationships (McMillan and Woodruff, 1999), institutional establishment including the legal system and level of political links (Barrot, 2015) influence, among others, trade credit demand and supply. Simultaneously, trade credit channel is guided by the financial infrastructure of an economy (Fisman and Love, 2003; Miwa and Ramseyer, 2008; Degryse et al., 2018), economic policy uncertainty (D'Mello and Toscano, 2020; Jory et al., 2020), financial market development (Ge and Qiu, 2007; Abdulla et al., 2017), company financials (Andrieu et al., 2018; García-Teruel and Martínez-Solano, 2010b; Dary and James, 2019), financing constraints and macro-economic drivers (Bastos and Pindado, 2013; Jinjarak, 2015; Carbó-Valverde et al., 2016). Beside the quantitative perspectives, qualitative research shows that both cultural (El Ghoul and Zheng, 2016; Bedendo et al., 2020) and social factors (Wu et al., 2014; Levine et al., 2018) influence the demand and supply of trade credit. Within an increasingly complex global market, trade credit channel plays a critical role in driving business growth (Chowdhury and Lang, 1996; Fisman and Love, 2003) and affects inventory policy (Haley and Higgins, 1973; Chung et al., 2005), competitive advantage (Pirttilä et al., 2019), and firm performance (Allen et al., 2019).

For financial theory, trade credit is a financing agreement between non-financial corporations extended to meet the business objectives of firms with or without banking intermediation (Schwartz, 1974; Mian and Smith, 1992; García-Teruel and Martínez-Solano, 2010b). The agreement has a double advantage: the buyer has a financial benefit as it extends the payment deadline; those who sell have a commercial advantage because, thanks to the extension of the terms of collection, they can benefit from an increase in sales. Acknowledging trade credit as a significant source of finance to small and medium enterprises (SMEs), Petersen and Rajan (1997) empirically validate the financing and marketing theories of trade credit and provide some

of the methodologies widely followed in later empirical investigations (García-Teruel and Martínez-Solano, 2010a; Afrifa and Gyapong, 2017). The financing hypothesis dominates in rationalizing the empirical results given that the point at which commercial credit replaces bank credit as an alternative source of financing is among those most frequently dealt with (Bastos and Pindado, 2013; Carbò-Valverde et al., 2016; Norden et al., 2020).

Among the different dimensions of trade credit research, we aim at addressing the following research questions summarized in the hypotheses described in the next sections, with the objective to broaden the vision of the work on the financial perspective of the trade credit channel:

- Is there a significant relationship between trade credit and SME investment decisions?
- If so, does this relationship vary in provinces with an abundance of relationship cooperative banks' branches?
- If so, is their relationship affected by firm-bank relationship features?

3.1 Trade credit channel and firm investment decisions

The empirical evidence on the influence of the trade credit channel on firm investment decisions is still mixed. While Coricelli and Frigerio (2016) find that the provision of trade credit may drain the investment-supportive liquidity, Dass et al. (2015) provide theoretical and empirical evidence that the provision of trade credit can act as a commitment device for making relationship-specific investments⁵. Consistent with Coricelli and Frigerio (2016), Ferrando and Wolski (2018) study the relationship between net trade credit and firms' investment levels by focusing on both financially distressed firms and crisis periods by using a large panel of more than 10 million firms in 23 EU countries over the period 2004–2014. Their results suggest that net trade credit has an overall negative impact on firm investment due to its liquidity-absorbing nature. Moreover, Carbò-Valverde et al. (2016) examine whether trade credit provided an alternative source of external finance to SMEs during the crisis. Using firm-level Spanish data they document that credit-constrained SMEs depend on trade credit, but not bank loans, and that the intensity of this dependence increased during the financial crisis.

⁻

⁵ In an environment with incomplete contracts and bargaining power, Dass et al. (2015) argue that trade credit poses itself as a quality guarantee mechanism to mitigate the uncertainty of the downstream company on the quality of acquired goods. This quality can be enhanced by extra investment efforts of the upstream company. Therefore, their empirical investigation confirms this theoretical prediction by suggesting that trade credit, in the form of accounts receivables, is positively associated with relationship-specific investments.

Our above discussion on the effects of the trade credit channel on firm investment can be summarized in our first hypothesis (H1):

H1. The trade credit channel significantly affects corporate investment and, as such, brings over real effects in the economy.

3.2 Trade credit channel and the local banking system

As described in Section 2, the structure of the Italian local banking system is characterized by geographical heterogeneity. Its effect on the relationship between trade credit and firm investment is not clear a priori. On the one hand, a greater proportion of local relationship cooperative banks' branches in Italian provinces should facilitate access to bank loans and reduces finance constraints for SMEs with which they have established relationships in the operating area (Alessandrini et al., 2009; La Rocca et al., 2010). This leads us to expect that facilitated access to bank credit should weaken firm reliance on the trade credit channel to finance investments, if the given firm is located in a province with a high proportion of relationship cooperative banks' branches. Indeed, local cooperative banks benefit from a competitive advantage over nationwide banks in small business lending, since the latter are characterized by organizational complexity and face more severe communication frictions due to the greater distance between their headquarters and local branches (Berger et al., 2005b). On the other hand, a number of studies challenge this conventional paradigm according to which community banks (small, single-market, local institutions) form strong relationships with informationally opaque small businesses (Haynes et al., 1999; Cole et al., 2004; Scott, 2004; Berger et al., 2005b). These studies document that changes in lending technologies and deregulation of the banking industry has made it easier for large and nonlocal banks to serve small, opaque firms (Berger et al., 2014). Berger and Udell (2006) highlight that large banks are also capable of serving small and opaque firms well using hard-information technologies, i.e., credit scoring and lending against fixed asset collateral, consistent with Frame et al. (2001) and Berger et al. (2005a). de la Torre et al. (2010) find that both large and small banks cater to small firms with an increasing use of hard information-based technologies (Berger and Black, 2011; Berger et al. 2011). Both Frame et al. (2004) and DeYoung et al. (2011) document that small business credit scoring is accountable for an increase in lending distance.

Our second hypothesis (H2) summarizes the influence of the structure of the local banking system on the nexus between trade credit and firm investment decisions:

H2. The effect of the trade credit channel on firm investments is affected by the proportion of local cooperative banks' branches relying on soft information-intensive relationship banking,

3.3 Trade credit channel and relationship banking

Studies investigating the nexus between trade credit and relationship banking are limited. In this regard, a valuable contribution is provided by Giannetti et al. (2011) who document that "trade credit usage is correlated with the buyer's banking relationships", by focusing on the United States. Specifically, they show that firms receiving trade credit secure financing from relatively uninformed banks, thus supporting our findings that the proprietary soft information collected by the bank decreases the likelihood of the borrowing firm being credit-constrained (Berger and Udell, 2002; Gobbi and Sette, 2014; Presbitero et al., 2014; Bolton et al., 2016). Moreover, Giannetti et al. (2011) find that firms that make greater use of the trade credit channel have shorter relationships with their banks. In a similar vein, Von Thadden (1995) and Degryse and Ongena (2004) provide evidence that firms borrowing from distant banks for short periods are generally considered to have arm's length relations with their lenders who gather only limited, and mostly hard, information about their borrowing corporates. Furthermore, McMillan and Woodruff (1999), Johnson et al. (2002), and Uchida et al. (2007) document that longer duration of trading relationships is often associated with a greater use of the trade credit channel.

Our third hypothesis (H3) summarizes the influence of the trade credit channel on firm investment decisions according to relationship banking-related features:

H3. The effect of the trade credit channel on firm investments is decreasing in the length of the firm-bank relationships, while is unaffected by the bank-firm distance as cooperative banks operate with SMEs mostly on a local basis.

4. Data

This paper exploits the granularity and the uniqueness of a proprietary dataset combining public firm and bank-level financial information with private bank-firm lending information on a sample of 18,682 Italian SMEs operating with 99 cooperative banks over the time period 2008-2014.

Public financial information concerns the economic and financial characteristics of the companies and banks in our sample. The firm-level variables concern the composition of assets and liabilities, the intensity of the investment activity, the economic performance measured by the return on assets (ROA). The bank-level characteristics concern the profitability (i.e. return on

average asset - ROAA), capitalization (i.e., bank's equity ratio) and the quality of the loans disbursed (impaied loans over total loans) of the cooperative banks in our sample. Cooperative banks, which cover about 7.2% of the Italian loan market (McKillop et. al., 2020), are known to be close to the territory and attentive to the needs of smaller companies (Baccarani et al., 2013), such as the SMEs populating our dataset. We collect data on firm accounting variables from Centrale dei Bilanci (CEBI) and on bank variables from Bankscope⁶.

Private lending information is of two types. On the one hand, we observe the duration of the relationship between the company and the bank and the distance between the borrower and the bank. The data were provided exclusively to us by CSD, an Italian company that manages the information system of more than 100 Italian cooperative banks. On the other hand, they include data on a borrower's debt position towards the banking system that are collected from the Italian Credit Register (CR) managed by the Bank of Italy.

Our setting involves the presence of 99 cooperative banks located in 100 Italian provinces spread over the Italian territory and lending to 83 different industries according to the 2-digit Ateco industry classification⁷. In particular, firms in our sample belong to the following six macro-industries: agriculture, commerce, transports and hotels, manufacturing, building and services. We exclude public administration and financial firms. Companies are segmented on the basis of two main criteria: synthetic codes of economic activity, i.e., 2-digit Ateco industry classification, and their legal form. The Ateco code allows each company to be correctly associated with a specific sector, while the fact that all the companies in the sample belong to the capital companies segment makes it possible to have a homogeneous set of accounting data drawn from their annual financial statements.

Lastly, data on the number of cooperative banks' branches located in Italian provinces are collected from the Bank of Italy.

Table 1 reports the definition of the dependent and explanatory variables used in the analysis and some descriptive statistics.

[Insert table 1 about here]

Out of these 100 Italian provinces, 43 are located in Northern Italy, 21 in Central Italy, and 36 in Southern Italy and on the islands.

12

⁶ For SMEs, trade credit forms a substantial part of their balance sheet, i.e., in our sample, on average, firms engage in providing and receiving trade credit at the levels of 39.3% and 28.9% respectively, as a share of their total assets.

5. Empirical methodology

The empirical approach relies on panel data estimation on a sample of 18,682 Italian SMEs in the time period ranging from 2008 to 2014. Our panel data structure allows us to control for time invariant and unobserved factors specific to each firm-bank pair driving differences in firm investment decisions. All the estimated models are saturated by time and industry-specific fixed effects, or by a vector of industry-year fixed effects with industries characterized at the 2-digit level of the Ateco 2007 classification of economic activities⁸.

In model selection, we performed the Hausman test to determine whether to implement a fixed- or random-effects model. This test leads us to reject the null hypothesis of random effects, thereby accepting the implementation of a fixed-effect (FE) model. To mitigate the issue of the heteroskedasticity of residuals, detected by performing the modified Wald test for groupwise heteroskedasticity in FE regression models, we test all our models using heteroskedasticity-robust standard errors. To avoid the effect of outliers driving our results, data are winsorized at the 5% level⁹.

5.1 Trade credit channel and firm investment decisions

Firstly, we test whether the trade credit channel affects firm investment. Following Kaplan and Zingales (1997), Fazzari et al. (2000), and Carbo'-Valverde et al. (2016), firm investment is introduced as the ratio of the year-on-year change in capital expenditure relative to the total amount of capital in the previous year, where capital expenditure is computed as the annual change in fixed assets, i.e., inclusive of financial, intangible and tangible fixed assets, plus amortization and depreciation ($\Delta CAPEX_t$).

Following Afrifa and Gyapong (2016), the trade credit channel is introduced as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets, i.e., net trade credit, NTC^{10} . Specifically, the net trade credit position

_

⁸ This classification is the national version of the European nomenclature, Nace Rev. 2, published in the Official Journal of 20 December 2006 (Regulation (EC) no 1893/2006 of the European Parliament and of the Council of 20 December 2006).

⁹ Given that it is often not encouraged to reject outliers, especially when there is no tangible explanation about the occurrence of outliers as in the context of our paper, one remedy suggested by the related literature is to lower the impact of the outlying observations by implementing winsorization (Dixon & Tukey, 1969; Dixon & Yuen, 1974), a popular accommodation method to reduce the weights of outliers by replacing them with a specific percentile of data-dependent values (Dixon & Yuen, 1974; Orr et al., 1991). The percentile of winsorization is suggested to be adjusted according to the shape of the distribution (Dixon & Yuen, 1974; Tukey, 1962). Therefore, to minimize such influence, data are winsorized at the 5% level, i.e., 5% in each tail, following focused visual inspection of the distribution of our sample values in order to minimize the disproportionate influence of outliers on statistical analyses that might lead to distortions of statistical significance tests.

¹⁰ In unreported regressions, available upon request, we test the robustness of our results by scaling the difference between receivables and payables by the firms' total sales. This leaves our findings qualitatively and quantitatively unaffected.

measures the cash tied up in the trade cycle before it comes back out as cash again. The longer the net trade credit position of a given firm, the greater is its working capital requirement (Bernstein and Wild, 1998). On the one hand, a positive net trade credit position requires the company to finance the days taken by account receivables to be cashed in by tapping its financing sources. On the other hand, a negative net trade credit position reflects a situation in which the given firm is being paid for its sales before having to pay its account payables. Following Petersen and Rajan (1997), when we view the firm as a supplier, "its accounts receivable are a proxy for how much it lends its customers", while when we view the firm as a customer, "its accounts payable are its borrowing from its supplier". In this perspective, we construct a measure of the firm net trade credit position by examining commercial relationships cultivated by a firm in the role of both customer (borrower) and supplier (lender) according to the above definitions. Net trade credit is taken at time t rather than t-1 to mimic the shortterm nature of trade credit contracts, usually less than one year (Ferrando and Wolski, 2018). However, we recognize that it can be potentially susceptible to reverse causality bias and therefore we run instrumental variable estimation as a robustness test to address this issue as described in section 6.1.

Our fixed effects (FE) panel data baseline regression model for the trade credit channel and firm investment decisions takes the following form:

$$\Delta CAPEX_{i,n,t} = \alpha + \beta_1 NTC_{i,n,t} + \sum_{j=1}^k \delta_j X_{i,n,t} + \varphi_{i,k} + \gamma D_{industry}_n + \delta D_{year}_t + \mu D_{industry}_n * D_{year}_t + \varepsilon_{i,n,t}$$

$$\tag{1}$$

where $X_{i,n,t}$ is the vector of control variables representing firm-specific characteristics for firm i, operating in industry n in year t described in section 4.5; $D_{industry_n}$ are industry dummies to control for industry-specific effects according to the 2-digit Ateco industry classification; D_{year_t} are yearly time dummies to control for time-specific effects; and $\varepsilon_{i,t}$ is the error term for firm i in year t; $D_{industry_n} * D_{year_t}$ are industry-year fixed effects in order to control for all time-varying shocks at the industry level. Moreover, the nature of the panel dataset with fixed effects ($\varphi_{i,k}$) allows us to control for firm i-bank k pair-specific unobserved heterogeneity.

A description of the construction of the model variables, as well as their descriptive statistics, is presented in Table 1.

5.2 Trade credit channel and the local banking system

Secondly, we explore whether the structure of the local banking system affects the relationship between trade credit and SME investment decisions. That is, we explore whether a high proportion of cooperative banks' branches in a province should reduce the need for trade credit and, as such, moderate the relationship between trade credit and firm investment decisions. To this purpose, we interact our measure of net trade credit, i.e., *NTC*, with the variable *cooperative banking*, a binary variable equal to 1 if the density of cooperative banks' branches with respect to the number of firms operating in a given Italian province is above the median value of the distribution and 0 otherwise¹¹.

Our FE panel data regression model for the trade credit channel and the local banking system takes the following form:

$$\Delta CAPEX_{i,n,t} = \alpha + \beta_1 NTC_{i,n,t} + \beta_2 NTC_{i,n,t} * cooperative banking + \beta_3 cooperative banking +$$

$$\sum_{j=1}^k \delta_j X_{i,n,t} + \varphi_{i,k} + \gamma D_{industry}_n + \delta D_{year}_t + \mu D_{industry}_n * D_{year}_t + \varepsilon_{i,n,t}$$
 (2)

where $X_{i,n,t}$ is the vector of control variables representing firm-specific characteristics for firm i, operating in industry n in year t described in section 4.5.

By investigating the sign and the statistical significance of the coefficient β_2 in Eq. (2) we can make inferences on whether the structure of the local banking system affects firm reliance on the trade credit channel to spur investments. On the one hand, a positive and statistically significant coefficient β_2 would document that a high proportion of relationship cooperative banks' branches should facilitate small business access to bank credit and, as such, weaken firm reliance on the trade credit channel to spur investments. On the other hand, a non-statistically significant coefficient β_2 would signal that changes in lending technologies and deregulation of the banking industry have made it easier for large and nonlocal banks to serve small, opaque firms, thus eroding the comparative advantage of local soft information-intensive relationship banks in serving small businesses.

15

¹¹ In unreported regressions, available upon request, we adopt an alternative cutoff point to construct *cooperative* banking according to which we set the latter equal to 1 if the density of cooperative banks' branches with respect to the number of firms operating in a given Italian province is above the 75th percentile of the distribution and 0 otherwise. Our main results remain unchanged.

5.3 Trade credit channel and relationship banking

Thirdly, we investigate whether the influence of the trade credit channel on firm investment decisions is moderated by firm-bank relationship features¹². Specifically, we are interested in investigating whether the specific characteristics of a given firm-bank financing relationship leads to a differential impact of the trade credit channel on firm investments. To this purpose, we construct the binary variable *relationship banking* that is proxied by either the length of the bank-firm relationship (*length of relationship*) or the geographical distance between the borrowing firm and the lending branch (*borrower-to-branch distance*).

Our FE panel data regression model for the trade credit channel and relationship banking takes the following form:

$$\Delta CAPEX_{i,n,t} = \alpha + \beta_1 NTC_{i,n,t} + \beta_2 NTC_{i,n,t} * relationship banking + \beta_3 relationship banking +$$

$$\sum_{j=1}^k \delta_j X_{i,n,t} + \varphi_{i,k} + \gamma D_{industry_n} + \delta D_{year_t} + \mu D_{industry_n} * D_{year_t} + \varepsilon_{i,n,t}$$
(3)

where $X_{i,n,t}$ is the vector of control variables representing firm-specific characteristics for firm i, operating in industry n in year t described in section 4.5.

Therefore, at the firm-bank relationship level, we observe the length of the bank-firm relationship and the bank-borrower distance (Agarwal, 2010; Agarwal and Hauswald, 2010; Alessandrini et al., 2009; Carbo'-Valverde et al., 2016; Filomeni et al. 2020a; Filomeni et al. 2020b)¹³. The motivation lays in the fact that, a priori, firm characterized by deeper banking relationships might decrease their reliance on the trade credit channel as a funding source in a context where relationship banking provides benefits for both the lender and the borrower in terms of a Pareto-improving exchange of information between the parties involved and several welfare-improving contractual features.

due diligence during loan underwriting and performs subsequent monitoring of the borrower after the loan is

disbursed (Berger and Udell, 2006).

¹² In the study of Berger and Udell (2002), relationship banking is defined as a lending technology which is alternative to transaction banking based exclusively on hard information; in this respect, they affirm that: "the information gathered over time has significant value beyond the firm's financial statements, collateral, and credit score, helping the relationship lender deal with informational opacity problems better than potential transaction lenders". Relationship banking is based on soft information. In this respect, the figure of the loan officer is of crucial importance in collecting soft information since the loan officer not only establishes the firm-bank relationship in the first place, but also conducts

Agarwal (2010) and Agarwal and Hauswald (2010) make use of the binary variable repeated relationship in their empirical analysis, equal to 1 in there is a prior relationship, and 0 otherwise, as well as of the borrower-to-branch distance. Agarwal (2010) investigate the effect of physical firm-bank distance on the collection and use of subjective intelligence in informationally opaque credit markets. Agarwal and Hauswald (2010) examine how the allocation of authority inside the bank affects the production, transmission and strategic use of soft information in lending decisions. Agarwal (2010) show that greater bank-firm distance erodes the quality of local soft information, therefore reducing the reliance on soft information in credit decisions. Alessandrini et al. (2009) find that smaller operational distance does not always enhance credit availability.

Firstly, the length of the bank-firm relationship allows us to assess the nature and the strength of the bank-firm relationship, which facilitates the collection of borrower-specific information and is defined as a categorical variable *length of relationship* equal to number of years of the lending relationship. To test whether the influence of the trade credit channel on firm investment is negatively associated to the length of the bank-firm relationship, we interact NTC with *length of relationship* and investigate the sign and the significance of coefficient β_2 in Eq. (3) associated with the interaction term. As anticipated, *length of relationship* allows us to capture the nature and the strength of the bank-firm relationship which might significantly affect the relationship between trade credit and firm investment behavior.

Secondly, the bank-borrower distance is assessed by defining the binary variable borrower-to-branch distance equal to one if the borrower's headquarters and the branch in which the loan officer in charge of developing the firm-bank relationship are located in the same geographical location, and zero otherwise. Since relationship lending is considered an appropriate tool for bank lending to more informationally opaque SMEs, and evidence suggests that it could alleviate credit constraints (Berger and Udell, 2002; Scott, 2006; Gobbi and Sette, 2014; Presbitero et al., 2014; Bolton et al., 2016), therefore it potentially mitigates firm reliance on the trade credit channel as a financing source. To test whether the influence of the trade credit channel on firm investment significantly differs according to the bank-borrower distance, we interact NTC with borrower-to-branch distance and investigate the sign and the significance of coefficient β_2 in Eq. (3) associated with the interaction term. As anticipated, borrower-to-branch distance allows us to assess the ease of soft information transmission between the borrowing firm and the lending bank due to greater geographical proximity.

5.4 Control variables

In our regressions, we control for several firm-specific characteristics that could influence firm investment behaviour, other than the trade credit channel.

At the firm level, we control for the firm current ratio $(CR_{i,n,t})$, measured as the ratio of current assets over current liabilities, in order to supervise the effect that the hedging of current assets may have on firm investment changes. We also control for the firm unit labor cost $(ULC_{i,n,t})$ as labor costs may divert cash flows away from investment purposes, for the firm inventory period $(Inventory_{i,n,t})$ as the longer the inventory is held, the more the warehouse drain financial resources and reduces the firm investment possibilities, for the firm inventory to assets ratio $(Inventory/Total Assets_{i,n,t})$ which reflects the portion of assets tied up in inventory since the inventory changes are associated with the release of cash flows which

promote investments, and for the firm return on assets $(ROA_{i,n,t})$ to take into account the influence of firm profitability on investment decisions. Finally, we include three different additional ratios to control for the size of the borrowing firm, i.e., the firm tangible assets to total assets ratio (Tangible Assets/Total Assets_{i,n,t}), the firm intangible assets to total assets ratio (Intangible Assets/Total Assets_{i.n.t}), the firm financial fixed assets to total assets ratio (Financial Fixed Assets/Total Assets_{i,n,t}).

At the firm-bank level, we control for the short term and long term bank debt scaled by total liabilities (Bank Debt/Total Liabilities_{i,k,t}), in line with Coluzzi et al. (2012) and Heshmati (2001), who show that access to bank loans is an important driver of firm growth and may, consequently, affect firm investment behavior.

6. Results

Firstly, we are interested in investigating the relationship between the trade credit channel and firm investment decisions. The former is measured by the firm net trade credit position in given year t (NTC). The latter, our dependent variable, is defined as the ratio of the year-on-year change in capital expenditure between the previous year t-1 and the current quarter t relative to the total amount of capital in the previous year t-1 ($\Delta CAPEX$). Table 2 shows the results from running our baseline model, which is empirically defined in equation (1). We are interested in investigating the statistical significance of the coefficient β_1 associated with our main regressor (NTC) to analyse whether the trade credit channel significantly influences firm investment behavior. In this regard, columns (1) to (4) of Table 2 show that the NTC coefficient β_1 is negative and statistically significant at the 1% level. We interpret this result to mean that an increase in the firm net trade credit position is reflected in a decrease in the growth rate of capital expenditure by our sample of SMEs, ceteris paribus. Confirmatory graphical evidence is provided by Figure 1 showing the predicted outcomes of $\Delta CAPEX$ at different percentiles of NTC: when moving from the 10^{th} percentile (-19%), characterizing a firm receiving trade credit, to the 90th percentile of NTC distribution, reflecting a firm extending trade credit (46.8%), $\Delta CAPEX$ decreases, on average, from 22.3% to -2%, respectively. ¹⁴

[Insert table 2 about here]

¹⁴ The impact of explanatory variables on the growth rate of firm investment is computed using the "margins" command in Stata, keeping all the other variables at the average.

Secondly, we explore whether the structure of the local banking system affects the relationship between trade credit and SME investment decisions. We do this by exploiting within-country variation in the Italian local banking structure characterized by different degrees of inter-personal financing relationships. As described in Section 2, the Italian banking system is populated by a large number of cooperative banks that operate in restricted territorial areas, mostly located in the north part of the country (Alessandrini and Zazzaro, 1999; Alessandrini et al., 2009) and that benefit from competitive advantages over nationwide banks by more reliably processing soft information collected directly and indirectly through personal bank-firm relationships (Howorth and Moro, 2006). Indeed, nationwide banks are more afflicted by organizational complexity and communications frictions in hardening and transmitting soft information over greater distance between their headquarters and local branches (Berger et al., 2005; Filomeni et al., 2020a). This prompts them to research economies of scale in the processing of hard information and to specialize in transactional lending (Ferri and Neuberger, 2014). Model specification is shown by Equation (2). The estimation results are reported in columns (1)-(2) in Table 3. The sign and the statistical significance of the coefficient β_2 on the interaction terms NTC*cooperative banking in both column (1) and (2) is positive and statistically significant, thus suggesting that a high proportion of relationship cooperative banks' branches facilitates small business access to bank credit and, as such, weaken firm reliance on the trade credit channel to spur investments. This is supportive of the notion that local banks still benefit from a competitive advantage over nationwide banks in small business lending (Berger et al., 2005b), despite changes in lending technologies and deregulation of the banking industry are challenging their advantage to serve small, opaque firms (Berger et al., 2014)¹⁵.

[Insert table 3 about here]

Thirdly, we investigate whether the influence of the trade credit channel on firm investment decisions differs according to relationship banking features. Model specification is shown by Equation (3). Table 4 shows the results of this analysis.

First, to test whether the influence of the trade credit channel on firm investment significantly differs according to the length of the bank-firm relationship, we interact *NTC* with *length of relationship* which is equal to number of years of the lending relationship. Model specification is shown by Equation (3). The estimation results of our model are reported in

_

¹⁵ In unreported regressions, available upon request, we also control for the proportion of non-cooperative banks' branches located in a given province with respect to the number of firms operating in that province to account for the structure of the local banking system. This leaves our main results unaffected.

column (1) of Table 4. The sign and the statistical significance of the coefficient β_2 on the interaction term NTC*length of relationship is positive and statistically significant at the 1% and 5% level, as shown in columns (1) and (2) in Table 4, thus suggesting that longer bank-firm relationships are associated with a decreased influence of the trade credit channel on firm investment decisions. These results are supportive of the extant literature providing evidence that proprietary "soft" information, i.e., relationship banking, decreases the likelihood of the borrowing firm being credit-constrained (Berger and Udell, 2002; Scott, 2006; Gobbi and Sette, 2014; Presbitero et al., 2014; Bolton et al., 2016).

Second, to test whether the influence of the trade credit channel on firm investment significantly differs according to the bank-borrower distance, we interact NTC with borrower-to-branch distance. Model specification is shown by Equation (3). The estimation results of our model are reported in column (3) of Table 4. As expected and in line with the evidence provided by Alessandrini et al. (2009) on Italian firms, the sign and the statistical significance of the coefficient β_2 on the interaction term NTC*borrower-to-branch distance is not statistically significant meaning that smaller borrower-to-branch distance does not always enhance credit availability. This result is motivated by the nature of our sample of cooperative banks operating with SMEs mostly on a local basis. Nevertheless, the main effect on NTC remains in line with our previous results as its coefficient is still negative and statistically significant at the 1% level.

[Insert table 4 about here]

7. Robustness tests

To confirm our empirical results on the relationship between the trade credit channel and firm investment behavior, we perform several robustness checks that leave our previous findings unchanged.

7.1 Addressing potential endogeneity concerns

In this section we address the endogeneity associated with potential reverse causality issues and omitted variable bias resulting from the simultaneous specification of the trade credit variable and investment. Therefore, to assuage concerns about potential endogeneity issues that might affect our estimation results, we perform instrumental variable (IV) estimation with respect to our baseline model of Equation (1) by using banking-related instruments.

We use as instrumental variables the equity ratio (Equity Ratio), the non-performing loan ratio (NPL ratio), the return on average assets (ROAA) and the bank size expressed as the logarithm of total assets (Bank Size), following Storz et al. (2017), Ferrando and Wolski (2018)

and Norden et al. (2020). In this regard, we follow the argument that the situation of a financing bank should be unrelated to company's investment decisions, but may affect the degree of trade credit in the corporate sector (Ferrando and Wolski, 2018)¹⁶, assuming that banks are not active firms' investors playing a determinant role in its operating and financing decisions. Therefore, these instruments could be referred to as pure numbers that are likely to affect the firm net trade credit position without directly affecting our dependent variable represented by the annual percentage change in the firm's capital expenditure. The validity and importance of the instruments for the control variables are verified using a number of diagnostic tests reported at the bottom of Table 5, which reports the second stage of the regressions and the value of the coefficients and standard errors of the two instrumental variables in the first stage. Results are reported in Table 5. Our variable of interest is the firm net trade credit position (NTC_{int}). Consistent with our previous results in Table 2, instrumental variable estimation confirms that an increase in net trade credit negatively affects firm investment behavior. This effect is statistically significant at the 1% level in model specification shown in Table 5. Overall, the results estimated by performing an IV estimation are in line with our baseline model's main findings and the diagnostic tests do not specify any problems regarding the application of instruments used, thus providing a reliable robustness check to our main results.

[Insert table 5 about here]

7.2 Influence of crisis periods

As an additional robustness check, we now run our baseline regression model in Equation (1) on the entire sample period by removing from the analysis those years characterized by acute financial instability, i.e., 2008, 2009 and 2011, to investigate whether our previous results might be influenced by in-crisis periods. Indeed, the 2007 financial crisis and the 2011 European Sovereign Debt Crisis have hampered access to bank credit (Kayshap and Stein, 2000; Puri et al., 2011; Jiménez et al., 2012; Jiménez et al., 2014) with a disproportionately greater effect for private, more informationally opaque firms (De Young et al., 2015). The results are shown in the first column of Table 6. The effect of the trade credit channel on firm investment remains qualitatively unchanged, i.e., negative and statistically significant at the 5% and 10% level, as shown in columns (1) and (2) in Table 6.

¹⁶ A likely scenario is the one in which weak banks may be unable to provide sufficient credit to the corporate sector to finance firm investment projects. In such a scenario, the variation in the firm investment levels explained by the corresponding bank specific situation materializes through the corporate funding structure, which is related to the use of the trade credit channel.

[Insert table 6 about here]

7.3 Accounting for firm heterogeneity

We now verify whether our results might differ across firms of different collateralization levels since collateral plays an important role in the firm capability of raising external funding. In this regard, empirical evidence shows that higher-collateralized firms face a lower cost of debt and benefit from higher availability of external finance (Benmelech and Bergman, 2011). Thus, firms which are more collateralized are less likely to be financially constrained, thus affecting their reliance on the trade credit channel as a funding source. For this purpose, we created the binary variable high collateral that takes a value of 1 for those firms whose ratio of fixed tangible assets over total assets is above the median value of the distribution, and 0 otherwise. We split our data sample into two sub-samples of high- and low-collateralized SMEs and re-run our baseline model in Equation (1). The results are displayed in Table 7. Again, our previous results hold robust as they remain qualitatively unchanged, with a negative and statistically significant effect at the 5% and 1% level of the trade credit channel on firm investment for the two sub-samples of both high- and low-collateralized firms, reported in columns (2)-(3) and (4)-(5) in Table 7, respectively. However, as expected, firm reliance on the trade credit channel to boost investments is decreasing in the degree of firm collateralization due to increased access to bank credit, as documented by the positive and statistically coefficient associated with the interaction term NTC*high collateral reported in column (1) in Table 7. Indeed, Cerqueiro et al. (2016) show that collateral plays an important and positive role in the provision of lending, while Benmelech and Bergman (2011) find a negative relationship between collateral and the cost of external debt finance.

[Insert table 7 about here]

7.4 Longer trade credit period

As a last robustness test, due to the short-term nature of trade credit contracts, we now test whether our results might be driven by those firms characterized by a trade credit period exceeding 365 days. To this purpose, we now re-run our baseline model in Equation (1) by removing from the analysis SMEs having a trade credit period exceeding one year with both days sales (DSO) and days payable (DPO) outstanding over 365 days. The results are displayed in Table 8. Even in this case, our previous results remain unaffected, with a negative and statistically significant effect at the 1% level of the trade credit channel on firm investment.

[Insert table 8 about here]

7.5 Additional controls

To control for possible influence from extra firm-specific shocks, we augment the main model specification by additional bank-specific variables which may influence firm investment decisions other than the trade credit channel.

First, we re-perform our baseline model in Equation (1) by controlling for those bank-specific characteristics used as instrumental variables in our IV estimation setting depicted in section 6.1, i.e., the equity ratio (Equity Ratio), the non-performing loan ratio (NPL ratio), the return on average assets (ROAA) and the bank size expressed as the logarithm of total assets (Bank Size). The results are displayed in Table 9. This inclusion leaves our main findings unaffected.

[Insert table 9 about here]

Second, we re-perform our baseline model in Equation (1) by controlling for the borrowing firm's regional fixed effects to control for possible macroeconomic shocks at the regional or provincial level which may affect firm investment decisions. The results are displayed in Table 10. Our main findings remain, even in this case, qualitatively unchanged.

[Insert table 10 about here]

8. Conclusions

Besides bank lending, evidence has shown that trade credit can be considered to be the next most important source of SME external financing (OECD, 2014). Within this context, this paper provides a novel contribution to the literature by investigating the extent to which SME reliance on the trade credit channel to finance investment decisions is affected by the structure of the local banking system and relationship banking features. Specifically, we address this research questions by focusing on a sample of 18,682 SMEs operating with 99 cooperative banks over the time period 2008-2014 and by exploiting the geographical heterogeneity of the Italian banking market characterized by several provinces populated by an abundance of cooperative banks' branches mostly relying on soft information in their credit relationships.

Firstly, we find a significant influence of the trade credit channel on firm investment decisions, suggesting that trade credit significantly affects the growth rate of firm investment.

Secondly, we document that those SMEs located in Italian geographical provinces characterized by an abundance of cooperative banks' branches relying on soft information-intensive relationship banking are less dependent on trade credit to finance their investment decisions. This is supportive of the view that local relationship cooperative banks still have competitive advantages over nationwide banks in small business lending, since the latter are characterized by organizational complexity and face more severe communication frictions due to the greater distance between their headquarters and local branches. This result is of particular relevance in light of the technological progress and deregulation in the banking sector that pushes towards eroding the comparative advantage of local relationship banks in serving small businesses.

Lastly, we provide evidence that shorter firm-bank relationships lead to a greater dependence of companies on the trade credit channel to boost investments, while the influence of firm-bank geographical proximity on this nexus is not significant due to the local nature of SMEs-cooperative banks credit relationships nurtured predominantly on a local basis.

To conclude, our results suggest that the trade credit channel plays a significant influence on firm investment decisions and that the magnitude of this influence depends on the structure of the local banking market and on the intensity of relationship banking. Since investments play a crucial role to boost SMEs' economic recovery, the results of this paper contribute to the current academic and policy debates on safeguarding and preserving business continuity in the midst of the current Covid-19 crisis, which is likely to drive many businesses into bankruptcies. Given the profound implications of this Covid-19-induced pandemic, fostering a deep understanding of the real effects of firm financing sources is paramount to avoid bankruptcy as it can not only support financially distressed companies to benefit from policy measures aimed at preserving firms relationships along the supply chain, but also it can make firms more confident to invest under stressed scenarios.

References

- Abdulla, Y., V. A. Dang, A. Khurshed, 2017. "Stock Market Listing and the Use of Trade Credit: Evidence from Public and Private Firms". *Journal of Corporate* Finance, 46(C), 391-410.
- Afrifa, G. A., E. Gyapong, 2017. "Net trade credit: what are the determinants?". *International Journal of Managerial Finance*, 13(3), 246-266.
- Agarwal, S., 2010. "Distance and Private Information in Lending". Review of Financial Studies, 23(7), 2757-2788.
- Agarwal, S., R. Hauswald, 2010. "Authority and Information". AFA 2011 Denver Meetings Paper.
- Agostino, M., Trivieri, F., 2014. "Does trade credit play a signalling role? Some evidence from SMEs microdata". *Small Business Economics*, 42, 131–151.
- Alessandrini, P., Zazzaro, A., 1999. "A 'possibilist' approach to local financial systems and regional development: The Italian experience". In R. Martin (Ed.), *Money and the space economy* (pp. 71–92). New York: Wiley.
- Alessandrini, P., A. Presbitero, A. Zazzaro, 2009. "Banks, Distances and Firms' Financing Constraints". Review of Finance, 13(2), 261-307.
- Allen, F., M. Qian, J. Xie, 2019. "Understanding informal financing". *Journal of Financial Intermediation*, 39, 19-33.
- Almeida, H., M. Campello, B. Laranjeira, S. Weisbenner (2012), "Corporate debt maturity and the real effects of the 2007 credit crisis", *Critical Finance Review*, 1, 3-58.
- Andrieu, G., R. Staglianò, P. Zwan, 2018. "Bank debt and trade credit for SMEs in Europe: firm-, industry-, and country-level determinants". *Small Business Economics*, 51(1), 245-264.
- Angelini, P., Di Salvo, R., Ferri, G., 1998. "Availability and cost of credit for small businesses: Customer relationships and credit cooperatives". *Journal of Banking and Finance*, 22, 925–954.
- Baccarani, C., G.M. Golinelli, E. Rullani, 2013. "Le BCC banche di territorio". CEDAM, Padova.
- Bank of Italy, 2019, "The 2018 Annual Report", Bank of Italy, 31 May, available at: https://www.bancaditalia.it/pubblicazioni/relazione-annuale/2018/
- Barrot, J.N., 2015. "Trade Credit and Industry Dynamics: Evidence from Trucking Firms". *The Journal of Finance*, 71(5), 1975-2016.
- Bartoli, F., G. Ferri, P. Murro, Z. Rotondi, 2013. "SME financing and the choice of lending Technology in Italy: Complementarity or substitutability?" *Journal of Banking and Finance*, 37, 5476–5485.

- Bastos, R., J. Pindado, 2013. "Trade credit during a financial crisis: a panel data analysis". *Journal of Business Ressearch*, 66(5), 614-620.
- Beck, T., A. Demirguc-Kunt, 2006. "Small and medium-size enterprises: Access to finance as a growth constraint". *Journal of Banking and Finance*, 30(11), 2931-2943.
- Beck, T., H. Degryse, R., De Haas, N., Van Horen, 2018. "When arm's length is too far: Relationship banking over the credit cycle". *Journal of Financial Economics*, 127(1), 174-196.
- Bedendo, M., E. Garcia-Appendini, L. Siming, 2020. "Cultural Preferences and Firm Financing Choices". *Journal of Financial and Quantitative Analysis*, 55(3), 897-930.
- Benfratello, L., Schiantarelli, F., Sembenelli, A., 2008. "Banks and innovation: Microeconometric evidence on Italian firms". *Journal of Financial Economics*, 90, 197–217.
- Benmelech E., N. K. Bergman, 2011. "Bankruptcy and the Collateral Channel". *Journal of Finance*, 66(2), 337-378.
- Berger, A.N., Black, L.K., 2011. "Bank size, and small business finance". *Journal of Banking and Finance*, 35, 724–735.
- Berger A., G. Udell, 1995. "Relationship Lending and Lines of Credit in Small Business Finance", *Journal of Business*, 68, 351-381.
- Berger A., G. Udell, 1998. "The economics of small business finance: The roles of private equity and debt markets in the financial growth cycle". *Journal of Banking and Finance*, 22, 613-673.
- Berger, A., G. Udell, 2002. "Small business credit availability and relationship lending: the importance of bank organizational structure". *The Economic Journal*, 112, F32-F53.
- Berger, A., G. Udell, 2006. "A more complete conceptual framework for SME finance". *Journal of Banking and Finance*, 30, 2945–2966.
- Berger, A.N., Frame, W.S., Miller, N., 2005a. "Credit scoring and the availability, small business credit". *Journal of Money, Credit, and Banking*, 37, 191–222.
- Berger, A. N., Miller, N. H., Petersen, M. A., Rajan, R. G., Stein, J. C., 2005b. "Does function follow organizational form? Evidence from the lending practices of large and small banks". *Journal of Financial Economics*, 76, 237–269.
- Berger, A.N., Cowan, A.M., Frame, W.S., 2011. "The surprising use of credit scoring in small business lending by community banks and the attendant effects on credit availability, risk, and profitability". *Journal of Financial Services Research*, 39, 1–17.
- Berger, A., W. Goulding, T. Rice, 2014. "Do small businesses still prefer community banks?" *Journal of Banking and Finance*, 44, 264–278.

- Bernstein, L.A., J.J. Wild, 1998. "Financial Statement Analysis", Irwin Professional Publishing, ISBN: 0256267367, 832.
- Biais, B., Gollier, C., 1997. "Trade credit and credit rationing". Review of Financial Studies, 10, 903–937.
- Bolton, P., X. Freixas, L. Gambacorta, P.E. Mistrulli, 2016. "Relationship and transaction lending in a crisis". Review of Financial Studies 29(10), 2643-2676.
- Bonaccorsi di Patti, E., & Gobbi, G., 2001. "The changing structure of local credit markets: Are small business special?". *Journal of Banking & Finance*, 25, 2209–2237.
- Breza, E., A. Liberman, 2017. "Financial Contracting and Organizational Form: Evidence from the Regulation of Trade Credit". *The Journal of Finance*, 72, 291-324.
- Burkart, M., T. Ellingsen, 2004, "In-kind finance: A theory of trade credit". *The American Economic Review*, 94(3), 569-590.
- Canto-Cuevas, F-J., M-J. Palacin-Sanchez, F. Di Pietro, 2019. "Trade Credit as a Sustainable Resource during an SME's Life Cycle". *Sustainability*, 11, 670.
- Carbo'-Valverde, S., F. Rodriguez-Fernandez, G. Udell, 2016. "Trade Credit, the Financial Crisis, and SME Access to Finance". *Journal of Money, Credit and Banking*, 48(1), 113-143.
- Carey, M., S. Prowse, J. Rea, G. Udell, 1993, "The Economics of Private Placements: A New Look," *Financial Markets, Institutions and Instruments*, 2 (August).
- Chowdhury, S. D., J.R. Lang, 1996. "Turnaround in small firms: An assessment of efficiency strategies". *Journal of Business Research*, 36(2), 169-178.
- Chung, K.-J., S.K. Goyal, Y.-F. Huang, 2005. "The optimal inventory policies under permissible delay in payments depending on the ordering quantity". *International Journal of Production Economics*, 95 (2), 203-213.
- Cole, R.A., Goldberg, L.G., White, L.J., 2004. "Cookie-cutter versus character: the micro structure of small business lending by large and small banks". *Journal of Financial and Quantitative Analysis*, 39, 227–251.
- Coluzzi, C., A. Ferrando, C. Martínez-Carrascal (2012). "Financing obstacles and growth: An analysis for Euro Area non-financial firms". *European Journal of Finance*, 21, 1-18.
- Coricelli, F., M. Frigerio, 2016. "Inter-Enterprise Credit and Adjustment During Financial Crises: The Role of Firm Size", CEPR Discussion Papers, No. 11680.
- Coulibaly, B., H. Sapriza, A. Zlate, 2013. "Financial frictions, trade credit, and the 2008–09 global financial crisis". *International Review of Economics and Finance*, 26(C), 25-38.
- D'Mello, R., F. Toscano, 2020. "Economic policy uncertainty and short-term financing: The case of trade credit". *Journal of Corporate Finance*, 64, 101686.

- Dary, S., H. S. James, 2019. "Does investment in trade credit matter for profitability? Evidence from publicly listed agro-food firms," *Research in International Business and Finance*, 47(C), 237-250.
- Dass, N., R.K. Jayant, N. Vikram, 2015. "Trade Credit, Relationship-specific Investment, and Product Market Power". Review of Finance, 19 (5), 1867–1923.
- de la Torre, A., Martinez Peria, M.S., Schmukler, S.L., 2010. "Bank involvement with SMEs: beyond relationship lending". *Journal of Banking and Finance*, 34, 2280–2293.
- Degryse, H., S. Ongena, 2004. "The impact of competition on bank orientation" *CESifo Working Paper Series 1119*.
- Degryse, H., K. Matthews, T. Zhao, 2018. "SMEs and access to bank credit: Evidence on the regional propagation of the financial crisis in the UK". *Journal of Financial Stability*, 38(C), 53-70.
- Del Gaudio, B.L., C. Porzio, V. Verdoliva, 2018. "Trade credit, SMEs and short-run survivorship: What we know and what we would like to know". *Qualitative Research Financial Markets*, 10(4), 346-362.
- Demirgüç-Kunt, A., V. Maksimovic, 2002. "Funding growth in bank-based and market-based financial systems: evidence from firm-level data". *Journal of Financial Economics*, 65, 337-363.
- DeYoung, R., Frame, W.S., Glennon, D., Nigro, P., 2011. "The information revolution and small business lending: the missing evidence". *Journal of Financial Services Research*, 39, 19–33.
- DeYoung, R., A. Gron, G. Torna, A. Winton, 2015. "Risk overhang and loan portfolio decisions: Small business loan supply before and during the financial crisis: Risk overhang and loan portfolio decisions". *The Journal of Finance*, 70(6), 2451-2488.
- Dixon, W. J., Tukey, J. W., 1969. "Approximate behavior of the distribution of Winsorized t (trimming/Winsorization 2)". *Technometrics*, 10(1), 83-98.
- Dixon, W. J., Yuen, K. K., 1974. "Trimming and Winsorization: A review". *Statistische Hefte*, 15(2-3), 150-170.
- Duchin, R., O. Ozbas, B.A. Sensoy, 2010. "Costly external finance, corporate investment, and the subprime mortgage credit crisis" *Journal of Financial Economics*, 97(3), 418-435.
- El Ghoul, S., X. Zheng, 2016. "Trade credit provision and national culture". *Journal of Corporate Finance*, 41(C), 475-501.
- Elsas, R., 2005. "Empirical determinants of relationship lending". *Journal of Financial Intermediation*, 14, 32–57.

- Fabbri, D., F. Klapper, 2008. "Market Power and the Matching of Trade Credit Terms". World Bank Policy Research Working Paper, No. 4754.
- Fazzari, S.M., R.G. Hubbard, B.C. Petersen (2000), Investment-Cash Flow Sensitivities are Useful: A Comment on Kaplan and Zingales, *The Quarterly Journal of Economics*, 115 (2), 695-705.
- Ferrando, A., K. Mulier, 2013. "Do firms use the trade credit channel to manage growth?". *Journal of Banking and Finance*, 37(8), 3035-3046.
- Ferrando, A., M. Wolski, 2018. "Investment of financially distressed firms: The role of trade credit". EIB Working Papers, European Investment Bank (EIB), No 2018/04.
- Ferri, G., P. Kalmi, E. Kerola, 2014. "Does bank ownership affect lending behavior? Evidence from the Euro area". *Journal of Banking and Finance*, 48, 194-209.
- Ferri, G., G. Neuberger, 2014. "The banking regulatory bubble and how to get out of it". Rivista di Politica Economica, vol. CIII III, 39-69.
- Filomeni, S., G. Udell, A. Zazzaro, 2020a. "Hardening Soft Information: Does Organizational Distance Matter?". *The European Journal of Finance (forthcoming)*. doi: https://doi.org/10.1080/1351847X.2020.1857812.
- Filomeni S., G. Udell, A. Zazzaro, 2020b. "Communication frictions in banking organizations: Evidence from credit score lending". *Economics Letters* 195C, 109412.
- Fisman, R., I. Love, 2003. "Trade Credit, Financial Intermediary Development, and Industry Growth". *Journal of Finance*, 58(1), 353-374.
- Frame, W.S., Srinivasan, A., Woosley, L., 2001. "The effect of credit scoring on small business lending". *Journal of Money, Credit, and Banking*, 33, 813–825.
- Frame, W.S., Padhi, M., Woolsey, L., 2004. "The effect of credit scoring on small business lending in low- and moderate income areas". *Financial Review*, 39, 35–54.
- Garcia-Appendini, E., J. Montoriol-Garriga, 2013. "Firms as liquidity providers: evidence from the 2007–2008 financial crisis". *Journal of Financial Economics*, 109, 272–291.
- García-Teruel, P.J., P. Martínez-Solano, 2010a. "A dynamic approach to accounts receivable: A study for Spanish SMEs". *European Financial Management*, 16(3), 400-421.
- García-Teruel, P.J., P. Martínez-Solano, 2010b. "Determinants of trade credit: a comparative study of European SMEs". *International Small Business Journal*, 28(3), 215-233.
- Ge, Y., J. Qiu, 2007. "Financial development, bank discrimination and trade credit". *Journal of Banking and Finance*, 31(2), 513-530.
- Giannetti, M., M. Burkart, T. Ellingsen, 2011, "What you sell is what you lend?". Explaining trade credit contracts". Review of Financial Studies, 24(4), 1261-1298.

- Gobbi, G., E. Sette, 2014. "Do Firms Benefit from Concentrating their Borrowing? Evidence from the Great Recession". *Review of Finance*, 18 (2), 527–556.
- Gonçalves, A.B., R.F. Schiozer, H.H. Sheng, 2018. "Trade credit and product market power during a financial crisis". *Journal of Corporate Finance*, 49, 308-323.
- Guiso, L., Sapienza, P., & Zingales, L., 2004. "Does local financial development matter?". Quarterly Journal of Economics, 119, 929–969.
- Haynes, G.W., Ou, C., Berney, R., 1999. "Small business borrowing from large and small banks". In: Blanton, J.L., Williams, A., Rhine, S.L.W. (Eds.), Business Access to Capital and Credit. Federal Reserve System Research Conference, pp. 287–327.
- Haley, C.W., R.C. Higgins, 1973. "Inventory policy and trade credit financing". *Management Science*, 20(4), 464-471.
- Hermes, N., E. Kihanga, R. Lensink, C. Lutz, 2015. "The determinants of trade credit use: the case of the Tanzanian rice market". *Applied Economics*, 47(30), 3164-3174.
- Heshmati, A., 2001. "On the Growth of Micro and Small Firms: Evidence from Sweden". *Small Business Economics* 17, 213–228.
- Hill, M. D., G.W. Kelly, M.J. Highfield, 2010. "Net operating working capital behavior: a first look". *Financial Management*, 39(2), 783-805.
- Howorth, C., Moro, A., 2006. "Trust within entrepreneur bank relationships: Insights from Italy". *Entrepreneurship Theory and Practice*, 30, 495–517.
- International Labour Organization (2019), "Small matters Global evidence on the contribution to employment by the self-employed, micro-enterprises and SMEs", *ILO Publications Production Unit*, ISBN: 978-92-2-133991-5.
- Jacobson, T., E. Von Schedvin, 2015. "Trade Credit and the Propagation of Corporate Failure: An Empirical Analysis". Econometrica, 83, 1315-1371.
- Jiménez, G, S. Ongena, J.-L. Peydró, J. Saurina, 2012. "Credit Supply and Monetary Policy: Identifying the Bank Balance-Sheet Channel with Loan Applications". *American Economic Review*, 102(5), 2301-2326.
- Jimenez, G., S. Ongena, J.-L. Peydró, J. Saurina, 2014. "Hazardous times for monetary policy: what do twenty-three million bank loans say about the effects of monetary policy on credit risk-taking?". *Econometrica*, 82, 463-505.
- Jinjarak, Y., 2015. "Supply Chains, Global Financial Shocks and Firm Behaviour towards Liquidity Needs". *The World Economy*, 38 (3), 425-444.
- Johnson, S., J. McMillan, C. Woodruff, 2002. "Property Rights and Finance", *American Economic Review*, 92 (5), 1335-1356.

- Jory, S.R., H.D. Khieu, T.N. Ngo, H.V. Phan, 2020. "The influence of economic policy uncertainty on corporate trade credit and firm value". *Journal of Corporate Finance*, 64, 101671.
- Kaplan, S. N., L. Zingales, 1997. "Do Investment-Cash Flow Sensitivities Provide Useful Measures of Financing Constraints?". *The Quarterly Journal of Economics*, 112 (1), 169-215.
- Kashyap, A.K., J.C. Stein, 2000. "What do a million observations on banks say about the transmission of monetary policy?". *American Economic Review*, 90(3), 407-428.
- La Rocca, M., La Rocca, T., Cariola, A., 2010. "The influence of local institutional differences on the capital structure of SMEs: Evidence from Italy". *International Small Business Journal*, 28, 234–257.
- Levine, R., C. Lin, W. Xie, 2018. "Corporate Resilience to Banking Crises: The Roles of Trust and Trade Credit". *Journal of Financial and Quantitative Analysis*, 53(4), 1441-1477.
- Liberti, J. M., M.A. Petersen, 2019. "Information: Hard and Soft." *The Review of Corporate Finance Studies*, 8(1), 1–41.
- Lopez, J. S., M. C. Mazzilis, M. Aliberti, S. Grimaldi, 2019. "Mutualism and credit quality: good practice and good results. The case of Italian Cooperative Credit Banks". Quaderni di ricerca del Credito Cooperativo.
- Love, I., L.A. Preve, V. Sarria-Allende, 2007, "Trade credit and bank credit: evidence from recent financial crisis". *Journal of Financial Economics*, 83(2), 453-469.
- Martínez-Sola, C., García-Teruel, P.J., Martínez-Solano, P., 2014. "Trade credit and SME profitability". *Small Business Economics*, 42, 561–577.
- McGuinness, G., T. Hogan, R. Powell. 2018. "European trade credit use and SME survival". *Journal of Corporate Finance*, 49, 81-103.
- McMillan, J., C. Woodruff, 1999. "Interfirm Relationships and Informal Credit in Vietnam". Quarterly Journal of Economics, 98, 1285-1320.
- McKillop, D., D. French., B. Quinn, A.L. Sobiech., J.O.S. Wilson, 2020. "Cooperative financial institutions: A review of the literature". *International Review of Financial Analysis*, 71, 1-11.
- Meltzer, A. H., 1960. "Mercantile credit, monetary policy, and size of firms". *The Review of Economics and Statistics*, 42(4), 429-437.
- Mian, S. L., C. W. Smith, 1992. "Accounts receivable management policy: Theory and evidence". *The Journal of Finance*, 47(1), 169-200.
- Miwa, Y., J.M. Ramseyer, 2008. "The Implications of Trade Credit for Bank Monitoring: Suggestive Evidence from Japan". *Journal of Economics and Management Strategy*, 17, 317-343.

- Modigliani, F., M. Miller, 1958. "The cost of capital, corporate finance and the theory of investment". *The American Economic Review*, 261-297.
- Molina, C.A., L.A. Preve, 2012. "An empirical analysis of the effect of financial distress on trade credit". *Financial Management*, Spring, 187–205.
- Ng, C., R. Smith, J. Smith, 1999. "Evidence on the Determinants of Credit Terms Used in Inter-Firm Trade", *The Journal of Finance*, 54, 1109-1129.
- Nilsen, J., 2002. "Trade Credit and the Bank Lending Channel". *Journal of Money, Credit and Banking*, 34(1), 226-53.
- Niskanen, J., M., Niskanen, 2000. "The Determinants of Corporate Trade Credit Policies in a Bank-dominated Financial Environment: the Case of Finnish Small Firms". *European Financial Management*, 12, 81-102.
- Norden, L., G. Udell, T. Wang, 2020. "Do bank bailouts affect the provision of trade credit?". *Journal of Corporate Finance*, 60, 101522.
- OECD, 2014. "Financial Market Trends, SMEs and the Credit Crunch: Current Financing Difficulties, Policy Measures and a Review of Literature". OECD Journal, vol. 2013/2.
- Ogawa, K., Sterken, E., Tokutsu, I., 2013. "The trade credit channel revisited: evidence from micro data of Japanese small firms". *Small Business Economics*, 40, 101–118.
- Orr, J. M., Sackett, P. R., DuBois, C. L. Z., 1991. "Outlier detection and treatment in I/O psychology: A survey of researcher beliefs and an empirical illustration". *Personnel Psychology*, 44(3), 473-486.
- Pattnaik, D., S. Kumar, A. Vashishtha, 2020a. "Trade credit research before and after the global financial crisis of T 2008 A bibliometric overview". Research in International Business and Finance, 54.
- Pattnaik, D., S. Kumar, A. Vashishtha, 2020b. "Research on trade credit a systematic review and bibliometric analysis". *Qualitative Research in Financial Markets*, forthcoming.
- Paul, S., R. Boden, 2008, "The secret life of UK trade credit supply: Setting a new research agenda". *The British Accounting Review*, 40, 272-281.
- Petersen, M. A., R.G. Rajan, 1994. "The Benefits of Lending Relationships: Evidence from Small Business Data", *Journal of Finance*, 49, 1367-1400.
- Petersen, M. A., R.G. Rajan, 1997. "Trade credit: theories and evidence". *The Review of Financial Studies*, 10(3), 661-691.
- Pirttilä, M, V.M. Virolainen, L. Lind, T. Kärri, 2019. "Working capital management in the Russian automotive industry supply chain". *International Journal of Production Economics*, 221, 107474.

- Presbitero, A., G. Udell, A. Zazzaro, 2014. "The home bias and the credit crunch: A regional perspective". *Journal of Money, Credit and Banking*, 46(1), 53-85.
- Presbitero, A. F., A. Zazzaro, 2011. "Competition and relationship lending: Friends or foes?" *Journal of Financial Intermediation*, 20(3), 387–413.
- Puri, M., J. Rocholl, S. Steffen, 2011. "Global Retail Lending in the Aftermath of the U.S. Financial Crisis: Distinguishing between Supply and Demand Effects". *Journal of Financial Economics*, 100(3), 556-578.
- Schwartz, R. A., 1974. "An economic model of trade credit". *Journal of Financial and Quantitative Analysis*, 9(04), 643-657.
- Scott, J.A., 2004. "Small business and value of community financial institutions". *Journal of Financial Services Research*, 25, 207–230.
- Scott, J.A., 2006. "Loan officer turnover and credit availability for small firms". *Journal of Small Business Management*, 44, 544-562.
- Shenoy, J., R. Williams, 2017. "Trade credit and the joint effects of supplier and customer financial characteristics". *Journal of Financial Intermediation*, 29 (C), 68-80.
- Stein, J., 2002. "Information Production and Capital Allocation: Decentralized Versus Hierarchical Firms". *Journal of Finance*, 57(5), 1891-1921.
- Stiglitz, J., A., Weiss, 1981. "Credit Rationing in Markets with Imperfect Information" *American Economic Review*, 71(3), 93-410.
- Storz, M., Koetter, M., Setzer, R., A., Westphal, 2017. "Do We Want These Two to Tango? On Zombie Firms and Stressed Banks in Europe". ECB Working Paper No. 2104.
- Taketa, K., G. Udell 2007. "Lending Channels and Financial Shocks: The Case of Small and Medium-Sized Enterprise Trade Credit and the Japanese Banking Crisis". *Monetary and Economic Studies*, 25(2), 1-44.
- Tukey, J. W., 1962. "The future of data analysis". The Annals of Mathematical Statistics, 33(1), 1-67.
- Uchida, H., G. Udell, W. Watanabe, 2007. "Bank Size and Lending Relationships in Japan" NBER Working Papers Series, wp 13005.
- Uchida, H., G. Udell, N. Yamori, 2012. "Loan officers and relationship lending to SMEs", Journal of Financial Intermediation, 21, 97-122.
- Von Thadden, E.-L., 1995. "Long-Term Contracts, Short-Term Investment and Monitoring,". The Review of Economic Studies, 62(4), 557–575.
- Vaidya, R. R., 2011. "The Determinants of Trade Credit: Evidence from Indian manufacturing firms". *Modern Economy*, 2(5), 707-716.

Wu, W., M. Firth, O.M. Rui, 2014. "Trust and the provision of trade credit". *Journal of Banking and Finance*, 39(C), 146-159.

Figures and tables

Adjusted Predictions of Net Trade Credit with 95% CIs

Timear Description of Firm Investment Growth

Order Description of Firm Investment Growth

Order Description of Net Trade Credit with 95% CIs

Order Description of Net Trade Credit with 95% CIs

Net Trade Credit with 95% CIs

Order Description of Net Trade Credit with 95% CIs

Net Trade Credit with 95% CIs

Figure 1. Predicted outcomes of $\Delta CAPEX$ at percentiles of NTC

Notes: on the x-axis are reported the 10th, 25th, 50th, 75th, and 95th percentiles of NTC distribution.

Table 1. Variables' descriptions and descriptive statistics

Variable	Description	Obs	Mean	Std. Dev.	Min	Max		
Dependent Variable								
<i>∆CAPEX</i>	year-on-year change in	52,913	0.101473	0.427132	-0.4	1.5		
	capital expenditure							
	Firm-level Variables							
NTC	net trade credit	79,655	0.123351	0.238352	-0.3355856	0.5961828		
CR	firm current ratio	80,715	120.4969	97.69209	6.4	400		
ULC	firm unit labor cost	71,071	21.17461	16.24462	2.04	62.33		
Bank Debt/Total	short and long term bank	76,262	40.05312	24.64493	3.88	92.24		
Liabilities	debt over total liabilities							
Inventory Period	firm's inventory period	63,335	164.405	299.3785	1.78	1243.05		
Financial Fixed	ratio of the firm's financial	50,881	0.069644	0.126612	0.0003037	0.4853714		
Assets/Total	fixed assets to total assets							
Assets								
Intangible	ratio of the firm's	61,714	4.175309	6.366742	0.04	23.95		
Assets/Total	intangible assets to total							
Assets	assets							
Tangible	ratio of the bank's firm	76,856	25.34375	23.92339	0.71	80.71		
Assets/Total	tangible assets to total							
Assets	assets							
Inventory/Total	ratio of the firm's	66,260	26.77742	26.16904	0.49	90.23		
Assets	inventory to total assets							
Total Assets	firm total assets	87,527	5157.792	8127.442	68,000	31,449,000		
ROA	firm's return on assets	84,027	1.865897	7.70142	-14.14	22.41		
DPO	Firm's days payable	79,999	177.6105	179.2663	24.54629	771.1868		
	outstanding							
DSO	Firm's days sales	80,516	179.8311	156.6654	17.69697	673.3621		
	outstanding							
	Relatio	onship Ban	king Variab	<u>lles</u>				
Length of	number of years of the	54,017	6.864172	6.967365	0	54		
Relationship	bank-firm relationship	[54,017]	[1.66431]	[0.95776]	[0]	[4.0073]		
·	[logarithm of 1 + length	[, -, -,]	[,	[5.5.5, 7.5]	[~]	[]		
	of relationship (years)]							
	or remaining (years)							

Borrower-to-branch	binary variable = 1 if the	52,569	0.648329	0.477497	0	1
distance	borrower and the bank are	,				
	located in the same					
	geographical location, and					
	0 otherwise					
	0 00000					
	<u>B</u>	ank-level V	Variables			
Bank ROAA	bank's return on average	82,196	0.211137	0.658931	-2.854	1.475
	assets					
Bank Equity Ratio	bank's equity ratio	82,196	9.262379	2.438896	2.416	19.017
Bank	ratio of the bank's	79,676	9.089273	5.855901	1.434	40.668
Impaired/Total	impaired loans over gross					
Loans	loans					
Bank Total Assets	Log of the bank's total	82,196	14.00308	0.723207	11.05406	16.02128
	assets					
Cooperative banking	Binary variable equal to 1	54,045	0.49	0.50	0	1
	if the density of	[54,045]	[0.002]	[0.001]	[0]	[0.009]
	cooperative banks'					
	branches measured by the					
	number of cooperative					
	branches located in a					
	province over the number					
	of firms operating in that					
	province is above the					
	median value of the					
	distribution, and 0					
	otherwise					
	[density of cooperative					
	banks' branches]					

Table 2. Baseline Model of Trade Credit

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	FE Panel	FE Panel	FE Panel	FE Panel	FE Panel	FE Panel
	Model	Model	Model	Model	Model	Model
NTC	-0.525***	-0.546***	-0.517***	-0.298***	-0.370***	-0.365***
	(0.054)	(0.054)	(0.054)	(0.053)	(0.072)	(0.071)
CR	,	,	,	,	-0.000	-0.000
					(0.000)	(0.000)
ULC					-0.005***	-0.005***
020					(0.001)	(0.001)
Bank Debt/Total					-0.000	-0.000
Liabilities					0.000	0.000
Luvillics					(0.001)	(0.001)
Inventory Period					0.000	0.000
111001101 / 1 011001					(0.000)	(0.000)
Financial Fixed/Total				1.276***	1.215***	1.208***
				1.2/0	1.213	1.200
Assets				(0.150)	(0.169)	(0.156)
T., +				(0.150) 0.021***	(0.168) 0.020***	(0.156) 0.022***
Intangible				0.021	0.020	0.022
Assets/Total Assets				(0,000)	(0.000)	(0,000)
T 11 4 . /T . 1				(0.002)	(0.002)	(0.002)
Tangible Assets/Total				0.014***	0.013***	0.014***
Assets				(0.00.1)		
				(0.001)	(0.001)	(0.001)
Inventory/Total Assets					-0.004***	-0.004***
					(0.001)	(0.001)
ROA					0.005***	0.005***
					(0.001)	(0.001)
Observations	16,240	16,240	16,240	16 240	16 240	16 240
		0.02	0.09	16,240 0.15	16,240 0.10	16,240
R-squared	0.04					0.16
Number of firm-	6,480	6,480	6,480	6,480	6,480	6,48 0
bank pairs	TTE C	170	NIC	110	1 7770	.
Yearly FE	YES	NO	NO	NO	YES	NO
Industry FE	NO	YES	NO	NO	YES	NO
Year x Industry FE	NO	NO	YES	YES	NO	YES

Note: The table presents the results of the FE panel regression analysis for the baselines model where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Columns (1) to (3) are a reduced form of the full model of Equation (1). Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. *** p<0.01, ** p<0.05, * p<0.1, respectively.

Table 3. Trade Credit & Cooperative Banking

MADIADI DO	(1)	(2)
VARIABLES	FE Panel Model	FE Panel Model
NTC	-0.479***	-0.477***
	(0.087)	(0.085)
NTC *cooperative banking	0.192**	0.197**
•	(0.082)	(0.080)
Cooperative banking	-0.005	-0.006
	(0.026)	(0.026)
CR	-0.000	-0.000
	(0.000)	(0.000)
ULC	-0.005***	-0.005***
	(0.001)	(0.001)
Bank Debt/Total Liabilities	-0.000	-0.000
	(0.001)	(0.001)
Inventory Period	0.000	0.000
	(0.000)	(0.000)
Financial Fixed/Total Assets	1.220***	1.213***
	(0.168)	(0.156)
Intangible Assets/Total Assets	0.020***	0.022***
	(0.002)	(0.002)
Tangible Assets/Total Assets	0.013***	0.014***
	(0.001)	(0.001)
Inventory/Total Assets	-0.004***	-0.004***
	(0.001)	(0.001)
ROA	0.005***	0.005***
	(0.001)	(0.001)
Observations	16,136	16,136
R-squared	0.11	0.16
Number of firm-bank pairs	6,439	6,439
Yearly FE	YES	NO
Industry FE	YES	NO
Year x Industry FE	NO	YES

Note: The table presents the results of the FE panel regression analysis for Trade Credit & Local Banking System model where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. *** p<0.01, ** p<0.05, * p<0.1, respectively.

Table 4. Trade Credit & Relationship Banking

VARIABLES	(1) FE Panel Model	(2) FE Panel Model	(3) FE Panel Model	(4) FE Panel Model
	Wodel	Wodel	Model	Woder
NTC	-0.462***	-0.441***	-0.455***	-0.448***
	(0.091)	(0.091)	(0.102)	(0.099)
NTC * length of relationship	0.012**	0.010**	,	,
3 7 1	(0.006)	(0.006)		
Length of relationship	-0.121***	0.014*		
	(0.025)	(0.007)		
NTC * borrower-to-branch distance			0.123	0.122
			(0.110)	(0.109)
Borrower-to-branch distance			-0.255	-0.325
			(0.211)	(0.312)
CR	-0.000	-0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)
ULC	-0.005***	-0.005***	-0.005***	-0.005***
	(0.001)	(0.001)	(0.001)	(0.001)
Bank Debt/Total Liabilities	-0.000	-0.000	-0.000	-0.000
	(0.001)	(0.001)	(0.001)	(0.001)
Inventory Period	0.000	0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)
Financial Fixed/Total Assets	1.214***	1.197***	1.212***	1.206***
T 11 1 10 10 1 1	(0.168)	(0.156)	(0.172)	(0.161)
Intangible Assets/Total Assets	0.020***	0.022***	0.020***	0.022***
T 21 4 1/T 1 4 1	(0.002)	(0.002)	(0.002)	(0.002)
Tangible Assets/Total Assets	0.013***	0.014***	0.013***	0.013***
I /T / I /	(0.001)	(0.001)	(0.001)	(0.001)
Inventory/Total Assets	-0.004***	-0.004***	-0.004***	-0.004***
DO 4	(0.001) 0.005***	(0.001) 0.004***	(0.001) 0.005***	(0.001) 0.005***
ROA				
	(0.001)	(0.001)	(0.001)	(0.001)
Observations	16,079	16,079	15,492	15,492
R-squared	0.10	0.16	0.11	0.16
Number of firm-bank pairs	6,429	6,429	6,154	6,154
Yearly FE	YES	NO	YES	NO
Industry FE	YES	NO	YES	NO
Year x Industry FE Note: The table presents the results of the F	NO	YES	NO	YES

Note: The table presents the results of the FE panel regression analysis for Trade Credit & Banking Relationships model where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. *** p<0.01, ** p<0.05, * p<0.1, respectively.

Table 5. IV Estimation: Baseline model

VARIABLES	IV Estimation
NTC	-6.197***
	(0.709)
CR	0.004***
	(0.001)
ULC	0.003
	(0.002)
Bank Debt/Total Liabilities	0.018***
Innerton Davied	(0.002) -0.000
Inventory Period	(0.000)
Financial Fixed/Total Assets	-1.243***
1 11000000 1 10000 1 10000 2 155005	(0.422)
Intangible Assets/Total Assets	-0.008
	(0.005)
Tangible Assets/Total Assets	-0.014***
	(0.004)
Inventory/Total Assets	-0.040***
	(0.005)
ROA	0.021***
	(0.003)
Observations	15,405
Number of firm-bank pairs	5,116
	,
Tests	26.460
Cragg-Donald Wald F statistic	36.169
Kleibergen-Paap rk Wald F statistic	32.295
Kleibergen-Paap Anderson-Rubin	0.0000 0.0000
Stock-Wright	0.0000
Hansen J	0.1570
Transcri y	0.1370
<u>First Stage</u>	
Equity Ratio	0.005***
ATEL E	(0.001)
NPL Ratio	0.002***
DO 4.4	(0.001)
ROAA	-0.001
Dank Cina	(0.001)
Bank Size	0.055***
	(0.009)

Note: The table reports in column (1) and (2) coefficient estimates and robust standard errors (in parentheses) for the two-stage treatment effects model of Equation (1). In column (1) we assume that all control variables used in the model are endogenous and we use their own lagged values as instruments, while in column (2) we treat NTC as endogenous and we use banking-related instruments. The sample period is 2008-2014. The first stage includes all explanatory variables in the second stage. The dependent variable is $\Delta CAPEX_{l,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{l,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. The Cragg-Donald Wald F statistic and the Kleibergen-Paap rk Wald F statistic represent weak identification tests. The Kleibergen-Paap is a test of under-identification distributed as chi-square under the null of under-identification. The Anderson Rubin and Stock-Wright LM S statistic are weak-instrument-robust inference tests, distributed as F-test and chi-square respectively, under the null that coefficients of the endogenous regressors in the structural equation are jointly equal to zero, and the over-identifying restrictions are valid. The Hansen J statistic is a test of the over-identifying restrictions, distributed as chi-square under the null of instrument validity. The first-stage Kleibergen-Paap rk Wald F statistic is a test for weak instrument. All variables are defined in Table 1. In the margin, we report coefficients and standard errors for the instrumental variables. Robust errors are reported in parentheses. **** p<0.01, *** p<0.05, * p<0.1, respectively.

Table 6. Trade Credit in Off-Crisis Periods

	(1)	(2)	
VARIABLES	FE Panel Model	FE Panel Model	
NTC	-0.282***	-0.230**	
	(0.104)	(0.102)	
CR	-0.000	-0.000	
	(0.000)	(0.000)	
ULC	-0.007***	-0.007***	
	(0.001)	(0.001)	
Bank Debt/Total Liabilities	-0.00Ó	-0.000	
•	(0.001)	(0.001)	
Inventory Period	0.000**	0.000**	
3	(0.000)	(0.000)	
Financial Fixed/Total Assets	0.957***	0.960***	
	(0.232)	(0.212)	
Intangible Assets/Total Assets	0.022***	0.023***	
G	(0.003)	(0.003)	
Tangible Assets/Total Assets	0.012***	0.012***	
	(0.001)	(0.001)	
Inventory/Total Assets	-0.004**	-0.004***	
	(0.002)	(0.002)	
ROA	0.002	0.001	
	(0.002)	(0.002)	
Observations	10,728	10,728	
Number of firm-bank pairs	0.09	0.14	
R-squared	5,798	5,798	
Yearly FE	YES	NO	
Industry FE	YES	NO	
Year x Industry FE	NO	YES	

Note: The table presents the results of the FE panel regression analysis for Trade Credit in Off-Crisis Periods model where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. **** p<0.01, *** p<0.05, ** p<0.1, respectively.

Table 7. Baseline Model: Low vs High Collateralised Firms

	(1)	(2)	(3)	(4)	(5)
VARIABLES	All Firms	` '	` '	High-Collate	
· · · · · · · · · · · · · · · · · · ·	1 100 1 00000	200 000000		1118/3 3011111	
NTC	-0.404***	-0.315***	-0.227**	-0.475***	-0.468***
	(0.078)	(0.105)	(0.103)	(0.109)	(0.109)
NTC * High Collateral	0.131**	,	,	()	,
0	(0.075)				
High Collateral	0.048**				
3	(0.024)				
CR	-0.000	-0.000	-0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
ULC	-0.005***	-0.005***	-0.005***	-0.004***	-0.004**
	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)
Bank Debt/Total Liabilities	-0.000	0.000	-0.000	-0.001	-0.001
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Inventory Period	0.000	0.000	0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Financial Fixed/Total Assets	1.217***	1.401***	1.535***	0.641***	0.582***
	(0.157)	(0.269)	(0.262)	(0.240)	(0.221)
Intangible Assets/Total Assets	0.021***	0.032***	0.035***	0.004	0.003
	(0.002)	(0.003)	(0.003)	(0.004)	(0.004)
Tangible Assets/Total Assets	0.012***	0.036***	0.039***	0.006***	0.006***
	(0.001)	(0.003)	(0.003)	(0.002)	(0.002)
Inventory/Total Assets	-0.004***	-0.002	-0.002	-0.007***	-0.008***
	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)
ROA	0.005***	0.002	0.001	0.005**	0.004***
	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)
Observations	16,240	7,806	7,806	8,434	8,434
R-squared	0.16	0.11	0.18	0.13	0.21
Number of firm-bank pairs	6,480	3,490	3,490	3,548	3,548
Yearly FE	NO	YES	NO	YES	NO
Industry FE	NO	YES	NO	YES	NO
Year x Industry FE	YES	NO	YES	NO	YES

Note: The table presents the results of the FE panel regression analysis for the Baseline Model: Low vs High Collateralized Firms where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. *** p<0.01, ** p<0.05, * p<0.1, respectively.

Table 8. Trade Credit & Trade Credit Period

-	(1)	(2)
VARIABLES	FE Panel Model	FE Panel Model
NTC	-0.361***	-0.369***
	(0.075)	(0.075)
CR	-0.000	-0.000
	(0.000)	(0.000)
ULC	-0.006***	-0.006***
	(0.001)	(0.002)
Bank Debt/Total Liabilities	-0.000	-0.000
	(0.001)	(0.001)
Inventory Period	0.000	0.000
	(0.000)	(0.000)
Financial Fixed/Total Assets	1.272***	1.245***
	(0.181)	(0.166)
Intangible Assets/Total Assets	0.020***	0.022***
	(0.003)	(0.002)
Tangible Assets/Total Assets	0.013***	0.014***
	(0.001)	(0.001)
Inventory/Total Assets	-0.004***	-0.005***
Ţ	(0.001)	(0.001)
ROA	0.004***	0.004***
	(0.001)	(0.001)
Observations	15,143	15,143
Number of firm-bank pairs	0.10	0.15
R-squared	6,117	6,117
Yearly FE	YES	NO
Industry FE	YES	NO
Year x Industry FE	NO	YES

Note: The table presents the results of the FE panel regression analysis for the Trade Credit & Trade Credit Period model where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. *** p<0.01, ** p<0.05, * p<0.1, respectively.

Table 9. Baseline Model of Trade Credit with Bank-Specific Controls

TIADIADI EG	(1)	(2)
VARIABLES	FE Panel Model	FE Panel Model
NTC	-0.301***	-0.300***
1110	(0.076)	(0.074)
CR	-0.000	-0.000
	(0.000)	(0.000)
ULC	-0.005***	-0.006***
323	(0.001)	(0.001)
Bank Debt/Total Liabilities	-0.001	-0.001
	(0.001)	(0.001)
Inventory Period	0.000	0.000
	(0.000)	(0.000)
Financial Fixed/Total Assets	1.362***	1.333***
,	(0.179)	(0.167)
Intangible Assets/Total Assets	0.022***	0.023***
······································	(0.003)	(0.003)
Tangible Assets/Total Assets	0.014***	0.014***
0 ,	(0.001)	(0.001)
Inventory/Total Assets	-0.003**	-0.004***
<i>J.</i>	(0.001)	(0.001)
ROA	0.004***	0.004***
	(0.001)	(0.001)
ROAA	0.004	0.004
	(0.008)	(0.008)
Equity Ratio	0.003	0.002
	(0.009)	(0.009)
NPL Ratio	0.002	0.003
	(0.003)	(0.003)
Bank Size	0.049	0.065
·	(0.089)	(0.093)
Observations	14,595	14,595
Number of firm-bank pairs	0.11	0.16
R-squared	5,986	5,986
Yearly FE	YES	NO
Industry FE	YES	NO
Year x Industry FE	NO	YES

Note: The table presents the results of the FE panel regression analysis for the Baseline Model with Bank-Specific Controls where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. *** p<0.01, ** p<0.05, * p<0.1, respectively.

Table 10. Baseline Model of Trade Credit with Borrower's Regional Fixed Effects

	(1)	(2)
VARIABLES	FE Panel Model	FE Panel Model
NTC	-0.366***	-0.361***
	(0.073)	(0.072)
CR	-0.000	-0.000
	(0.000)	(0.000)
ULC	-0.005***	-0.005***
	(0.001)	(0.001)
Bank Debt/Total Liabilities	-0.000	-0.000
	(0.001)	(0.001)
Inventory Period	0.000	0.000
	(0.000)	(0.000)
Financial Fixed/Total Assets	1.210***	1.204***
	(0.168)	(0.157)
Intangible Assets/Total Assets	0.020***	0.022***
	(0.002)	(0.002)
Tangible Assets/Total Assets	0.013***	0.014***
	(0.001)	(0.001)
Inventory/Total Assets	-0.003***	-0.004***
	(0.001)	(0.001)
ROA	0.005***	0.005***
	(0.001)	(0.001)
Observations	16,136	16,136
Number of firm-bank pairs	0.10	0.16
R-squared	6,439	6,439
Yearly FE	YES	NO
Industry FE	YES	NO
Year x Industry FE	NO	YES
Regional FE	YES	YES

Note: The table presents the results of the RE panel regression analysis for the Baseline Model with Borrower's Regional and Provincial Fixed Effects where the dependent variable is $\Delta CAPEX_{i,n,t}$ which is the annual percentage change in the firm's capital expenditure. $NTC_{i,n,t}$ represents the trade credit channel measured as the difference between the firm's account receivables and account payables scaled by the firm's size as measured by total assets. All variables are defined in Table 1. Year, Industry and Year*Industry fixed effects are incorporated in regressions where indicated (not reported). Robust errors are reported in parentheses. **** p<0.01, *** p<0.05, ** p<0.1, respectively.